

# 

**Citation:** Hernandez JR, Liu S, Fredregill CL, Pietrantonio PV (2023) Impact of the V410L *kdr* mutation and co-occurring genotypes at *kdr* sites 1016 and 1534 in the VGSC on the probability of survival of the mosquito *Aedes aegypti* (L.) to Permanone in Harris County, TX, USA. PLoS NegI Trop Dis 17(1): e0011033. https://doi.org/10.1371/ journal.pntd.0011033

Editor: Abdallah M. Samy, Faculty of Science, Ain Shams University (ASU), EGYPT

Received: July 22, 2022

Accepted: December 17, 2022

Published: January 23, 2023

**Copyright:** © 2023 Hernandez et al. This is an open access article distributed under the terms of the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

**Data Availability Statement:** All relevant data are within the manuscript and its Supporting information files.

Funding: We acknowledge the funding from the Western Gulf Center of Excellence for Vector-Borne Diseases (WGCVBD) [URL:https://www.utmb.edu/ wgcvbd] and the University of Texas Medical Branch at Galveston (US) [URL:https://www. utsystem.edu/institutions/university-of-texasRESEARCH ARTICLE

Impact of the V410L *kdr* mutation and cooccurring genotypes at *kdr* sites 1016 and 1534 in the VGSC on the probability of survival of the mosquito *Aedes aegypti* (L.) to Permanone in Harris County, TX, USA

#### Jonathan R. Hernandez<sup>1</sup>, Shuling Liu<sup>2<sup>#</sup></sup>, Chris L. Fredregill<sup>3</sup>, Patricia V. Pietrantonio<sup>1\*</sup>

 Department of Entomology, Texas A&M University, College Station, Texas, United States of America,
Department of Statistics, Texas A&M University, College Station, Texas, United States of America, 3 Harris County Public Health, Mosquito and Vector Control Division (HCPH-MVCD), Houston, Texas, United States of America

¤ Current address: Department of Family Medicine, Oregon Health and Sciences University, Portland, Oregon, United States of America

\* p-pietrantonio@tamu.edu

# Abstract

Harris County, TX, is the third most populous county in the USA and upon detection of arboviruses Harris County Public Health applies insecticides (e.g., pyrethroid-based Permanone 31-66) against adults of Culex quinquefasciatus to prevent disease transmission. Populations of Aedes aegypti, while not yet a target of public health control, are likely affected by pyrethroid exposure. As this species is a vector of emerging arboviruses, its resistance status to Permanone and the kdr mutations in the voltage-gated sodium channel (VGSC) associated with pyrethroid resistance were investigated. We examined females of known genotype at the V1016I and F1534C sites (N = 716) for their genotype at the 410 amino acid position in the VGSC, and for the influence of their kdr genotype on survival to Permanone at three different distances from the insecticide source in field tests. Most females (81.8%) had at least one resistant L allele at the 410 position, being the first report of the V410L mutation in Ae. aegypti for Texas. When only genotypes at the 410 position were analyzed, the LL genotype exhibited higher survivorship than VL or VV. Out of 27 possible tri-locus kdr genotypes only 23 were found. Analyses of the probability of survival of tri-locus genotypes and for the V410L genotype using a multivariate logistic regression model including area, distance, and genotype found significant interactions between distance and genotype. When only the most common tri-locus genotypes were analyzed (LL/II/CC, 48.2%; VL/II/ CC, 19.1%; and VV/II/CC, 10.1%) genotype had no effect on survival, but significant interactions of distance and genotype were found. This indicated that the V410L kdr allele increased survival probability at certain distances. Genotypes did not differ in survivorship at 7.62-m, but LL/II/CC had higher survivorship than VL/II/CC at 15.24- and 22.86-m. The model also identified differences in survivorship among the operational areas investigated.

medicalbranch-at-galveston] [grant number: 1U01CK000512], subcontract to P.V.P. through Texas A&M AgriLife Research. P.V.P. additionally acknowledges funding from Texas A&M AgriLife Research Insect Vector Diseases Grant Program [124196-77110]. P.V.P acknowledges USDA-NIFA for support of Hatch project number TEX0-2-9206. The funders had no role in study design, data collection and analysis, decision to publish, or preparation of the manuscript.

**Competing interests:** The authors have declared that no competing interests exist.

# Author summary

Preventing *Aedes aegypti* populations from spreading arboviruses in highly populated areas primarily relies upon chemical control with insecticides, such as pyrethroids, applied by public health agencies to target the female mosquito vector. However, the development of insecticide resistance in mosquito populations jeopardizes these vector control strategies. The impact of the different *kdr* mutations conferring resistance to pyrethroids on the survival of *Ae. aegypti* in the field is not fully understood. We previously found that two knockdown resistance (*kdr*) mutations in the voltage gated sodium channel (VGSC) did not increase survivorship in field cage tests despite their widespread presence and high frequency. In this paper, we report the first detection of the V410L *kdr* mutation in combination with the V1016I and F1534C genotypes. Most females were detected with *kdr* mutations at one, two or three of these sites in the VGSC. Genotyping females after insecticide application in field cage tests showed that the V410L mutations were associated with higher survivorship only at the 15.24- and 22.86-meter distances from the source of the spray application, as we observed that the LL/II/CC genotype had higher survivorship than the VL/II/CC genotype at these distances.

# Introduction

Aedes aegypti (L.) is an efficient primary vector for some of the most medically important arboviruses worldwide, including Zika (ZIKV), dengue (DENV), chikungunya (CHIKV), and yellow fever (YFV) in tropical and subtropical regions of the world [1,2]. The increased distribution of Ae. aegypti globally exacerbates the risk of arboviral transmission [3,4]. Considered a domestic mosquito, Ae. aegypti are capable of living in human dwellings, while also preferring human bloodmeals [5], which contributes to increasing arbovirus dissemination [6]. Human arboviral outbreaks caused by Ae. aegypti have resulted in millions of deaths [7,8]. This species is a competent urban vector for Mayaro virus, an emerging public health concern in South America [9,10], and is also a competent vector for the emerging zoonotic disease Rift Valley fever [11]. Therefore, the need to control of Ae. aegypti females has led to a reliance on the application of adulticides as one of the major strategies to suppress arboviral disease outbreaks [12]. The most widely synthetic adulticides applied, both by public vector control programs and homeowners are pyrethroids. These act on the central nervous system by modulating the voltage-gated sodium channel (VGSC) preventing channel closure, resulting in insect paralysis and death [13]. Excessive use of pyrethroid space sprays against disease vectors has produced a selection pressure that has caused the development of resistance in mosquito populations worldwide [14-16].

Target site mutations in the VGSC, known as knockdown resistance or *kdr*, represent the most studied mechanism of pyrethroid resistance in insects [17]. The most frequent *kdr* mutations in *Ae. aegypti* in the Americas are V1016I and F1534C, which are normally associated with each other [18–21]. A study using a congenic strain (1534C:ROCK) of *Ae. aegypti* homozygous resistant only at the F1534C position found that this genotype conferred resistance to both type I and type II pyrethroids [22].

In agreement with this finding, heterologous expression of the F1534C homozygous resistant mutant AaNav1-1 VGSC in *Xenopus* oocytes supports the role of the F1534C mutation in pyrethroid resistance [23]. In contrast, the homozygous resistant V1016I cRNA similarly expressed in the *Xenopus* system did not influence channel sensitivity to type I or II

pyrethroids. However, when both resistant genotypes were expressed simultaneously in oocytes ( $V^{1016}I+F^{1534}C$ ) the V1016I genotype provided an increased reduction in sensitivity to type I and II pyrethroids compared to the F1534C mutant ( $F^{1534}C$ ) alone [24]. A similar result was found *in vivo*, as heterozygous females at the 1016 and 1534 sites (e.g., VI/FC), resulting from the cross of a double homozygous resistant and a susceptible strain, had intermediate pyrethroid resistance [25].

An additional mutation in the Ae. aegypti VGSC at the amino acid residue position 410 from valine to leucine (V410L) was reported in 2017 [26], which has since then been reported in combination with the V1016I and F1534C kdr genotypes at high frequencies in the Americas [27,28]. However, there are conflicting reports on the influence of the V410L mutation in Ae. aegypti resistance to pyrethroids. In vitro electrophysiological assays support that the V410L kdr genotype alone reduces sensitivity to type I and type II pyrethroids [26]. The reduced sensitivity to permethrin was further enhanced by the presence of the F1534C resistant genotype, suggesting an additive effect of the two mutations on permethrin resistance [26]. In contrast, female mosquitoes of the congenic LKR strain (a resistant strain from La Mesa backcrossed into the pyrethroid susceptible Rockefeller strain) that was triple homozygous resistant for the V410L, V1016I, and F1534C kdr genotypes did not confer higher levels of resistance to seven insecticides compared to the resistance levels of 1534C:ROCK, a congenic strain with only the F1534C resistant genotype [22,29]. Resistance to these insecticides (cyhalothrin, cypermethrin, etofenprox, DDT, permethrin, cis-permethrin, and trans-permethrin) ranged from 7- to 16- fold when compared to the susceptible Rockefeller strain [22,29]. Harris County, TX, which includes the City of Houston, is the third-most populous county in the United States with over 4.73 million residents reported in 2020 [30]. The county's humid, subtropical climate lies within the ideal habitat range of Ae. aegypti [1]. Houston is one of the most vulnerable areas for flooding events, especially the densely populated downtown area [31]. Concerns about flooding in Harris County have increased after natural disasters such as Hurricane Harvey in 2017. Flooding in the Harris County area could increase the number of mosquito oviposition sites, which would increase the area at risk of exposure to mosquito-borne diseases [32, 33]. This rapidly growing metropolitan area thus requires the Harris County Public Health, Mosquito and Vector Control Division (HCPH-MVCD) to provide public health services to protect country residents from arbovirus outbreaks. HCPH-MVCD conducts yearlong surveillance and testing for several arboviruses transmitted by mosquitoes, including West Nile and St. Louis encephalitis viruses as well as DENV, CHIKV, and ZIKV [34].

Upon detection of arboviruses in one of their Mosquito and Vector Control Operational Areas (MVCOA), HCPH-MVCD schedules ground-based ultra-low volume (ULV) applications to reduce vector mosquito populations. These treatments have primarily been against *Culex quinquefasciatus* upon detection of West Nile virus [35]. Adult females of *Cx. quinque-fasciatus* are controlled by alternating ULV applications of the pyrethroid Permanone 31–66 (31% permethrin; 66% piperonyl butoxide (a synergist cytochrome P450 enzymes inhibitor)) and the organophosphate malathion utilizing HCPH-MVCD trucks mounted with London Fog 18–20 ULV sprayers that are able to reach effective spray distances up to 30.48 m [35]. While HCPH-MVCD has not yet targeted either *Ae. aegypti* or *Aedes albopictus* for control with synthetic pesticides, these species would be targeted if diseases such as those caused by ZIKV, DENV, or CHIKV viruses were detected within Harris County. If targeted application against *Aedes* mosquitoes were necessary, chemical interventions are planned to be performed with Permanone 31–66 using ULV handheld sprayers, each capable of reaching areas that trucks cannot, such as backyards. In addition, piperonyl butoxide is the only metabolic enzyme inhibitor found in formulated products with the active ingredient permethrin available for public health use. ULV handheld foggers used by HCPH-MVCD can reach distances up to 15.24 m and can be utilized indoors and outdoors [36]. Although *Ae. aegypti* mosquitoes have not been targeted for control, the frequent pyrethroid applications against *Culex quinquefasciatus* could have contributed to resistance development as both species are present in Harris County and these *Ae. aegypti* populations carry *kdr* mutations [21,37].

Previously, we analyzed in *Aedes aegypti* populations of Harris County the frequency of the two most reported *kdr* mutations in the Americas that had been associated with permethrin resistance, V1016I and F1534C [21]. These *kdr* mutations were widely distributed across Harris County, with 77% of all female mosquitoes analyzed carrying the double homozygous resistant genotype II/CC. While the field-collected mosquitoes exhibited increased survivorship (at three distances and up to 22.86 m) compared to the susceptible Orlando-strain females, there were no differences in survivorship among the different genotypes detected at the 1016 and 1534 residue sites. These results indicate that there may be other *kdr* genotypes that could explain the survivorship of the females in the field tests. The resistant V410L genotype in combination with V1016I and F1534C *kdr* genotypes has been detected in the United States, with the triple resistant genotype (LL/II/CC) being associated with permethrin resistance [27,38].

This present study builds upon earlier work by investigating the presence of the V410L mutation in Harris County and determining if it contributes to in-field resistance when occurring in mosquitoes with known V1016I and F1534C genotypes. Herein we report the first detection of the resistant V410L mutation for Texas and the increased survivorship of homozygous tri-locus genotypes.

# Materials and methods

#### Mosquitoes

The *Ae. aegypti* Orlando insecticide susceptible strain from Florida, U.S., originated in 1952 and since then has been maintained as a laboratory colony, which is used as a standard for baseline toxicological testing in this species [39, 40]. At the HCPH-MVCD in Harris County this strain has been maintained since 2015.

For egg field collection and rearing details, refer to Hernandez et al., 2021 [21]. Egg collection was conducted between November 2017 and August 2019, and the areas of origin of the mosquitoes and test dates were: Area 23, November 9, 2017; Area 419, September 9, 2018; Area 53, October 11, 2018; Area 73, November 6, 2018; Areas 45 and 75, July 16, 2019; Areas 601 and 806, August 6, 2019.

For field bioassays *Ae. aegypti* females were two-to-five-day-old either  $F_0$  generation obtained from eggs collected from the field, or from the laboratory Orlando-strain colony [21].

Genomic DNA used in this study was prepared from 716 females of *Ae. aegypti* that had been genotyped for the V1016I and F1534C sites in a previous study from our laboratory [21].

Briefly, DNA was isolated from individual mosquitoes using the ZYMO Quick-DNA Microprep kit, following a modified version of the manufacturer's protocol with a one-hour incubation period in the DNA lysis buffer that was used to obtain greater DNA yields. The concentration and quality of each DNA sample were determined using an automatic microplate reader (Tecan Infinite M200). Genomic DNA was diluted to 10 ng/ $\mu$ L for genomic DNA analyses.

## Field cage tests

Field cage tests that simulated real mosquito control scenarios were used to detect susceptibility of *Ae. aegypti* to Permanone 31–66 at three distances were described in Hernandez et al. [21]. Briefly, the field test plot utilized a 3 x 3 design with 3 rows of three-cage-carrying posts positioned at 7.62-, 15.24-, and 22.86-m downwind of the insecticide application source. Within each distance, posts were positioned at 7.62-m from each other as subsamples. Each post had at least three cages, with at least two cages containing female mosquitoes from the field and one cage with pyrethroid-susceptible Orlando-strain females. Three similar posts with cages were placed upwind from the treatment source as negative controls (expected to survive), and to correct the mortality in the treated cages, if needed [41]. All tests used the maximum application rate for permethrin of 0.008 kg a.i./ha. Each single pass test had an operator applying Permanone 31–66 with a Colt-4 ULV Handheld Fogger Sprayer (London Fog, Minneapolis, MN). The direction of the pesticide treatment was perpendicular to the walking path [21].

#### Genomic DNA analyses

To detect the kdr genotypes in the VGSC at the 410 position (V410L) in females from the field tests, we used a DNA melting curve analysis of allele-specific PCR products carried out in 96-well plates using a QuantStudio 6 Pro system (Applied Biosystems, Foster City, CA) and diagnostic primers for the V410L mutations, as described by Saavedra et al. [20]. Peaks were analyzed using the Thermo Fisher's Design and Analysis v 2.3 software (Thermo Fisher Scientific Inc, Waltham, MA, US). For the V410L mutation detection, the 10 µL reaction mixture contained 5 µL of PowerUP SYBR Green Master Mix (Thermo Fisher Scientific, MA), 0.2 µM of the V410fw and 410rev primers (Table 1), 0.17 µM of the L410fw (Table 1), and 1 µL (10 ng) of gDNA template. PCR conditions consisted of a 3 min 95°C activation step, followed by 35 cycles at 95°C for 10 s, 60°C for 10 s, and 72°C for 30 s, with a final extension step of 15 s at 95°C. The melting curves were determined from 65°C to 95°C, with an increase of 0.2°C every 1 s. Based on the melting curves, a single peak at 80°C corresponded to the 113 bp product of the homozygous resistant mutant (LL), a peak at 83°C corresponded to the 133 bp product of the homozygous susceptible wild-type (VV), and two peaks, one at each of both temperatures corresponded to a heterozygous mutant (VL). Each PCR plate included i) a positive control to detect the wild-type susceptible genotype (VV) using the gDNA of one Orlando strain female and primers V410fw and 410rev, ii) one negative control that used the same primers and nuclease-free water instead of genomic DNA. All samples were analyzed in triplicate.

## Statistical analyses

Statistical analyses were performed using R software (R version 4.0.2, R Foundation for Statistical Computing, Vienna, Austria) and GraphPad Prism (San Diego, CA). Frequency and the survival rate of female mosquitoes were calculated for all subgroups of each predictor including genotype, distance, area, and post position. A Pearson Chi-Square test was used to test the univariate association between the survivorship and each individual predictor. The Fisher exact test was used for pairwise comparisons. Multivariate logistic regression models were fitted to assess the relationship between survival rate and all the predictors simultaneously. First,

Table 1. Sequences of primers used for detection of the *kdr* V410L in the *Aedes aegypti* VGSC. Primers feature base pair mismatches introduced at the third base from 3' end to increase allele specificity (italics); the diagnostic differential nucleotide is in bold, underlined [42].

Kdr Site	Primer Name	Primer Sequence (5' $\rightarrow$ 3')			
V410L	V410fw	GCGGGCAGGGCGGGGGGGGGGGGGGGCCATCTTCTTGGGTTCGTTC			
	L410fw	GCGGGCATCTTCTTGGGTTCGTTCTACCAT <u>T</u>			
	410rev	TTCTTCCTCGGCGGCCTCTT			

https://doi.org/10.1371/journal.pntd.0011033.t001

due to the sparsity of data, including low counts of certain genotypes and/or at specific distances, we used the Least Absolute Shrinkage and Selection Operator (LASSO) logistic regression model to select the most correlated predictors for survival rate [43]. The LASSO technique is widely used with sparse data since it can improve the quality of predictions by shrinking regression coefficients compared to models estimated with regular unpenalized maximum likelihood approaches which may tend to overfit the data. Some coefficients are set to zero based on LASSO, and therefore parsimony is also achieved for the selected model. The final model included genotype, distance, area and the interaction between genotype and distance. Second, since the confidence interval estimates using the LASSO method can be quite complicated, one common practice is to do statistical inference based on the selected model. Hence, we refitted the logistic model by excluding subgroups of the combinations of genotypes and distances with less than or equal to 16 females (the sample size of 16 corresponds to a margin of error of 0.25) or those subgroups with LASSO coefficients equal to zero. Next, we evaluated the goodness-of-fit using pseudo-R square and area under the curve (AUC) to evaluate our final model performance. Lastly, sensitivity analysis was performed by fitting the model using full data or using other thresholds of low count genotypes. We compared the model fitting results including parameter coefficients and goodness-of-fit to validate our final model. A multiple comparison adjustment was applied to control for type I errors when comparing more than two groups. The odds ratio of each predictor was calculated and visualized using a forest plot. P-values less than 0.05 were considered statistically significant.

The impact of the factors, area, genotype, and distance on the probability of survival of the mosquitoes was assessed through a logistic regression model. The model of the log-odds, where p equals probability of survival, is:

 $\log\left(\frac{p}{1-p}\right) = \beta_0 + \beta_1 * \text{Genotype} + \beta_2 * \text{Distance} + \beta_3 * \text{Area} + \beta_4 * \text{Genotype} * \text{Distance}$ 

# Results

## Genotype analysis

To determine the influence of a third *kdr* mutation in the VGSC on the survivorship of females tested in the Harris County's field cage tests, 716 *Ae. aegypti* females previously individually analyzed for the V1016I and F1534C *kdr* mutations were genotyped for the 410 position, and the results are shown in Fig 1 and Table 2. This V410L mutation was widespread across all eight areas analyzed and present at high frequency, with 54.1% of the females (N = 387) possessing the homozygous resistant LL genotype. Across the eight areas, the presence of the LL genotype ranged from 38.5% in area 23 to 96.4% in area 75 (Fig 1 and Table 2). When compared to the homozygous resistant genotypes at the V1016I and F1534C sites (II and CC), the resistant LL genotype at the 410 position was less frequent in seven of the eight operational areas, except area 45. Area 45 was unusual in that females had a higher frequency of the 410 LL genotype (77.8%) than either the resistant 1016 II (57.8%) or 1534 CC (64.4%) genotypes (Fig 1 and Table 2).

The tri-locus genotypes found were not equally distributed in all areas (Table 2 and Figs 1 and 2). Among the different operational areas, area 53 had the highest variety of genotypes observed, with seventeen genotypes detected within that area, followed by areas 23 and 73 with thirteen genotypes each, and area 601 with eleven different genotypes. Areas 45, 75, and 419 had the lowest variety of genotypes, with seven, two, and three, respectively (Table 2). The triple-*kdr* genotype (LL/II/CC) was the most abundant genotype observed in all areas, ranging



**Fig 1. Proportion of** *kdr* **mutations in female mosquitoes shown by operational control area in Harris County.** The pie charts represent the genotypic frequencies for each of the *kdr* mutations detected in individual females of *Ae. aegypti* collected for the eight operational areas in Harris County, adapted from Hernandez et al. [21]. Proportions of the genotypes at the 410 site are in shades of yellow (LL, VL, VV), and those for the 1016 (II, VI, VV) and 1534 (CC, FC, FF) sites previously genotyped are represented by shades of red and blue, respectively. In each pie chart, the lightest color intensity represents the wild-type, susceptible genotypes (410 VV; 1016 VV; or 1534 FF) and the darkest color indicates the resistant genotype (410 LL; 1016 IL; 1534 CC), with the intermediate color intensities corresponding to the heterozygotes (410 VL; 1016 VI; 1534 FC). For each area, the proportion of tri-locus homozygous resistant (LL/II/CC) females is shown in black, while the proportion of all other tri-locus genotypes at the 410, 1016, and 1534 genotypes are in white. The number of genotypes per area is indicated on the corresponding color on each pie chart. Survivorship data (Survival %) for each area (shown in columns below pie charts) by distance from the spray application [1) 7.62, 2) 15.24, 3) 22.86, and 4) 38.1 m] was previously reported [21] but is shown here for clarity. Map source: edited in QGIS v.3.22.1 software (www.agis.org). The layers for county (Map service: Harris County boundary masked) (https://www.arcgis.com/home/item.html?id=87a4455991c146259cf6c2e2384e8b71) were created by Harris County Public Health and are publicly available. There are no special restrictions or limitations on the terms of use of the layers applied to this map.

https://doi.org/10.1371/journal.pntd.0011033.g001

from 35.9% in areas 23 and 53, to 96.42% in area 75 (Fig 1 and Table 2). This genotype comprised 48.2% of all females tested and the proportions of LL/II/CC genotypes among the eight different areas were statistically different (Fig 2; Chi-square test, *P*-value < 0.0001). This genotype was followed in frequency by VL/II/CC (19.1%), VV/II/CC (10.1%), VL/VI/FC (3.8%), and VL/VI/CC (3.4%) (Table 2). Variation was also observed for some of these most abundant genotypes, for example, VL/II/CC ranged from 0 in area 75 to 34.6% in area 601 and VV/II/ CC ranged from 0 in area 419 to 21.8% in area 73. Eleven other tri-locus genotypes (VV/VV/ FF, VV/VV/FC, VV/VV/CC, VV/VI/FC, VV/VI/CC, VL/VV/FF, VL/VV/CC, LL/VV/CC, LL/ VI/FC, LL/VI/CC, and LL/II/FF) ranged between 0.3% and 2.8% of all females tested. The remaining seven tri-locus genotypes, VV/II/FC, VL/VV/FC, VL/II/FF, VL/II/FC, LL/VV/FF, LL/VV/FC, and LL/II/FC each had only one female representative. These were detected in areas 23, 53, 73, and 806, likely due to high variety of observed genotypes within these areas and indicates the sampling of egg collections was representative of the population.

	Mosquito Control Operational Areas								
	23	45	53	73	75	419	601	806	
Genotypes	n; (%)	n; (%)	n; (%)	n; (%)	n; (%)	n; (%)	n; (%)	n; (%)	n; (%)
VV/VV/FF	1; (1.3)	0; (0)	4; (3.1)	2; (1.3)	0; (0)	1;(4.8)	0;(0)	0;(0)	8; (1.1)
VV/VV/FC	1;(1.3)	0;(0)	4;(3.1)	5; (3.2)	0;(0)	0;(0)	0;(0)	0;(0)	10; (1.4)
VV/VV/CC	2; (2.6)	0;(0)	6; (4.6)	8; (5.1)	0;(0)	0;(0)	4; (3.8)	0;(0)	20; (2.8)
VV/VI/FC	2; (2.6)	0;(0)	4; (3.1)	7; (4.5)	0;(0)	0;(0)	0;(0)	0;(0)	13; (1.8)
VV/VI/CC	0;(0)	0;(0)	2; (1.5)	2; (1.3)	0;(0)	0;(0)	2; (1.9)	0;(0)	6; (0.8)
VV/II/FC	0;(0)	0;(0)	0;(0)	0;(0)	0;(0)	0;(0)	0;(0)	1; (1.0)	1; (0.1)
VV/II/CC	17; (21.8)	1; (2.2)	14; (10.7)	23; (14.7)	3; (3.6)	0; (0)	1; (1.0)	13; (13.4)	72; (10.1)
VL/VV/FF	0; (0)	0; (0)	4; (3.1)	0; (0)	0; (0)	0; (0)	0; (0)	0; (0)	4; (0.6)
VL/VV/FC	0; (0)	0; (0)	1; (0.8)	0; (0)	0; (0)	0; (0)	0; (0)	0; (0)	1; (0.1)
VL/VV/CC	1; (1.3)	0; (0)	2; (1.5)	0; (0)	0; (0)	0; (0)	1; (1.0)	0; (0)	4; (0.6)
VL/VI/FC	5; (6.4)	5; (11.1)	4; (3.1)	9; (5.8)	0; (0)	0; (0)	3; (2.9)	1; (1.0)	27; (3.8)
VL/VI/CC	2; (2.6)	0; (0)	5; (3.8)	8; (5.1)	0; (0)	0; (0)	5; (4.8)	4; (4.1)	24; (3.4)
VL/II/FF	1; (1.3)	0; (0)	0; (0)	0; (0)	0; (0)	0; (0)	0; (0)	0; (0)	1; (0.1)
VL/II/FC	1; (1.3)	0; (0)	0; (0)	0; (0)	0; (0)	0; (0)	0; (0)	0; (0)	1; (0.1)
VL/II/CC	15; (19.2)	4; (8.9)	26; (19.8)	27; (17.3)	0; (0)	1; (4.8)	36; (34.6)	28; (28.9)	137;(19.1)
LL/VV/FF	0; (0)	0; (0)	1; (0.8)	0; (0)	0; (0)	0; (0)	0; (0)	0; (0)	1; (0.1)
LL/VV/FC	0; (0)	0; (0)	0; (0)	1; (0.6)	0; (0)	0; (0)	0; (0)	0; (0)	1; (0.1)
LL/VV/CC	0; (0)	2; (4.4)	2; (1.5)	3; (1.9)	0; (0)	0; (0)	3; (2.9)	1; (1.0)	11; (1.5)
LL/VI/FC	0; (0)	11;(24.4)	3; (2.3)	1; (0.6)	0; (0)	0; (0)	1; (1.0)	2; (2.1)	18; (2.5)
LL/VI/CC	0; (0)	1; (2.2)	2; (1.5)	0; (0)	0; (0)	0; (0)	3; (2.9)	2; (2.1)	8; (1.1)
LL/II/FF	2; (2.6)	0; (0)	0; (0)	0; (0)	0; (0)	0; (0)	0; (0)	0; (0)	2; (0.3)
LL/II/FC	0; (0)	0; (0)	0; (0)	0; (0)	0; (0)	0; (0)	0; (0)	1; (1.0)	1; (0.1)
LL/II/CC	28; (35.9)	21; (46.7)	47; (35.9)	60; (38.5)	81; (96.4)	19; (90.5)	45; (43.3)	44; (45.4)	345;(48.2)
Total	78	45	131	156	84	21	104	97	716

Table 2. Tri-locus VGSC genotypes at sites 410, 1016, and 1534 detected in Ae. aegypti females of Harris County by operational area.

https://doi.org/10.1371/journal.pntd.0011033.t002

Analyses of female mosquitoes for the V410L genotypes only supported that areas differ in the number of homozygous resistant (LL), homozygous susceptible (VV) or heterozygous resistant genotypes (VL) (S1 Fig; Chi square, P < 0.0001). The statistical analyses with Fisher Exact test revealed two groups of areas with similar proportions for the VV, VL and LL genotypes, as follows: areas 806, 601, 53, 73, and 23 (S1 Fig) had more VV and VL individuals compared to areas 75 and 419. Area 45 was intermediate between the two groups.

Females were then compared for the tri-locus homozygous *kdr* genotype (LL/II/CC) versus to all other tri-locus genotypes. All areas had more LL/II/CC individuals than any of the other genotypes detected (Fig 2 and Table 2; Chi square, P < 0.0001). Pairwise comparisons among the different areas for their similarity in the proportion of LL/II/CC genotypes revealed two groups, as follows: 1) areas 75 and 419, and 2) areas 45, 806, 601, 53, 73, and 23 (Fig 2). Group 1 had an overall higher proportion of LL/II/CC females compared to the other areas, with 90.5% and 96.4% of the females detected being the resistant genotype in areas 419 and 75, respectively. The proportion of LL/II/CC females from group 2 ranged from 35.9% in areas 23 and 53, to 46.7% in area 45; the area 45 was intermediate being not significantly different from the two groups.

#### Linkage disequilibrium

Linkage disequilibrium (LD) analyses between pairs of alleles at the three *kdr* loci coding for V410L and V1016I, V410L and F1534C, and V1016I and F1534C, respectively, were conducted



Fig 2. Percentage of tri-locus homozygous resistant genotypes at the 410, 1016 and 1534 sites (LL/II/CC) vs other tri-locus genotypes detected per area. Percentage of the tri-locus (LL/II/CC) *kdr* mutation in females of *Ae. aegypti*. Significant differences were detected among the proportion of LL/II/CC genotypes observed across all areas (Chi-square; P < 0.0001). Horizontal lines above bars indicate areas in which the percentage of genotypes are not significantly different as per pairwise comparison analyses (Fisher Exact test; P < 0.05). White numbers in the black bars represent the number of females of the LL/II/CC genotype and black numbers above the bars represent all other genotypes observed in each area.

https://doi.org/10.1371/journal.pntd.0011033.g002

for each area (S1–S3 Tables). Area 75 could not be analyzed for linkage disequilibrium due to reaching fixation for the resistant genotypes II at 1016 and CC at the 1534 site. When analyzing linkage disequilibrium between the V410L and V1016I sites, six areas, areas 23, 419, 53, 73, 601 and 806, had significant LD with  $X^2 P$  values < 0.05 and D' values ranging from 0.422–0.999 (S1 Table). This supports that these areas are in LD for the 410 and 1016 loci except area 45, which had a higher proportion of V410L *kdr* alleles compared to the V1016I *kdr* alleles (Fig 1).

LD analyses for the V410L and F1534C sites, found that only three areas, 419, 53, and 73, had significant LD with  $X^2 P$  values < 0.05 and D' values ranging from 0.493–0.999 (S2 Table), which is likely due to the increased distance between these loci which would support more crossover events. Finally, the LD analyses for the V1016I and F1534C sites, found that areas 23, 419, 53, 73, 45, 75, 601 and 801, had significant LD with  $X^2 P$  values < 0.001 and D' values ranging from 0.507–0.999 (S3 Table). The analyses indicate that when comparing alleles at the different loci pairs, V1016I and F1534C loci were in strong, but not perfect, linkage disequilibrium.

#### Impact of V410L genotype on survivorship

We then asked if the presence of the V410L genotypes had an overall impact on survivorship regardless of their area of origin or distance from the treatment source (Fig 3).

After adjusting for differences in area and distance, there was a significant difference in the survivorship of mosquitoes among the three genotypes detected (Fisher's exact test, P = 0.004;



**Fig 3. Impact of genotype at the 410 site on female survival after Permanone 31–66 tests.** Estimated proportion of surviving females of *Ae. aegypti* by genotypes at the 410 site. There were highly significant differences in the proportion of surviving females among VV, VL and LL genotypes (P = 0.0037). The histogram shows the estimated proportions of surviving females ± CI, and different letters (a-b) above bars indicate differences in the proportions of surviving females, and the numbers above the histogram represent the number of dead females.

https://doi.org/10.1371/journal.pntd.0011033.g003

Fig 3). The survival proportion for LL was significantly higher than the survival proportions for VL and VV (Fisher's Exact Test, P = 0.044 and 0.0134, respectively). The survival proportions for VL and VV were not significantly different (Fisher's Exact Test, P = 0.074; Fig 3).

We then attempted to estimate the effect of genotype on survivorship at every distance. Survivorship of each tri-locus genotype at every distance tested also could not be statistically evaluated, due to the lack of representative genotypes at different distances from the sprayer in all areas studied (S4 Table). We thus analyzed for the V410L site (independently of genotypes at 1016 and 1534) the genotype effect on survivorship for each distance, but for all areas combined (S2 Fig). At 7.62 m, there were no differences in survivorship among genotypes (S2A Fig; Chi-square, P = 0.0516) from the Permanone 31–66 application. Genotypes at the 410 position showed differences in survivorship at 15.24 m (S2B Fig, Chi-square, P = 0.0001) and at 22.86 m (S2C Fig, Chi-square, P = 0.0001) with the homozygous resistant, LL, having higher survivorship. At 38.1 m, the test did not detect significant differences in survivorship for the three genotypes (S2D Fig; Chi-square, P = 0.8878).

# Effect of the most common tri-locus kdr genotypes on survivorship by distance

We then analyzed the survivorship of the three most common genotypes (VV/II/CC, VL/II/CC, LL/II/CC) compared to the susceptible Orlando strain controls on the same cage post at

Live

each distance. The choice for those three genotypes was due to the lack of representations of other genotypes when considering separation by different distances (S4 Table). The Orlando females had the highest mortality at each distance compared to the field collected females (Fig 4). At all distances the VV/II/CC genotype showed no significant difference to the VL/II/CC and LL/II/CC genotypes, while the latter two genotypes were significantly different from each other at each distance (Fig 4A-4C). The low number of the VV/II/CC (N = 60) females present may explain their similar survivorship to the VL/II/CC (N = 132) and LL/II/CC (N = 334) genotypes, as their low number were further stratified by distance. At the 7.62 m distance from the Permanone 31-66 source, survivorship of the field collected females ranged from 34.5% (LL/II/CC) to 56.5% (VL/II/CC) while the Orlando females only had 5% survival (Fig 4A), the lowest of all distances. The VL/II/CC genotype had significantly higher survivorship compared to the LL/II/CC genotype (Fisher's exact test, P = 0.0129; Fig 4A). However, the survivorship of the triple-resistant LL/II/CC was higher compared to the VL/II/CC genotype at 15.24 m (Fig 4B; Fisher's exact test, P = 0.0292) and 22.86 m distances (Fig 4C; Fisher's exact test, P < 0.0001) from the spray application. This analysis indicates that there are differences among the genotypes when compared at the different distances and the differences in mortality with respect to the Orlando strain were more pronounced at the furthest distance (Fig 4C).

#### Analysis of post position effect on survivorship

To determine if the differences in survivorship from the field cage tests may be attributed post position, the Orlando and field collected females with the three most common genotypes were analyzed by post number (S3 Fig). Post position refers to the placement of the posts within each distance, where post I is the closest to the beginning of the spraying and post III is the one closest to the ending of the spraying. At each post position, the Orlando females had the lowest survivorship compared to the field collected females. However, only at post position I, the VL/ II/CC and triple homozygous resistant genotype (LL/II/CC) had higher survivorship, being significantly different from the Orlando females, while the VV/II/CC genotype was intermediate (S3A Fig). At post positions II and III, the three field collected genotypes, VV/II/CC, VL/ II/CC, and LL/II/CC had significantly higher survivorship proportions compared the Orlando females (S3B and S3C Fig).

The most frequent genotypes VV/II/CC, VL/II/CC, and LL/II/CC were analyzed for survivorship differences among the different areas (S4 Fig). When areas 601 and 806 were not considered in the analyses due to their high survivorship, there were no significant differences in survivorship among areas for the LL/II/CC and VV/II/CC genotypes (S4A and S4C Fig), but females with the VL/II/CC genotype from area 53 (S4B Fig) had significantly lower survivorship compared to most areas. Area 53 also had the highest number of different genotypes detected in this study (S4 Fig and Table 2). Area 419 only had one female of genotype VL/II/CC and area 45 had 4 such females; statistically both areas, 419 and 45, were intermediate in survivorship between area 53 and areas 23 and 73; the latter two did not differ (S4 Fig). The results from this analysis support that areas influence survivorship even when females from different areas are of the same genotype, in this case VL/II/CC.

#### Impact of genotype on survivorship using logistic regression modeling

Logistic regression models for the 410 genotype alone and for the tri-locus genotypes (S4 Table) were fit to the data with interaction terms to determine the influence of genotypes on the survival of female mosquitoes. When evaluating the influence of the 410 genotype on survivorship with all data considered, genotype had a significant positive effect on survival (LR Chi-square, P = 0.002) (S4 Table, Panel A, top). The increase in survival of the 410 site genotype



**Fig 4. Influence of the most common tri-locus genotypes (410, 1016 and 1534) on the survivorship of females at each distance after Permanone 31–66 tests.** Panels show results of each of the tested distances from the Permanone 31–66 application source, as follows: (A) 7.62 m, (B) 15.24 m, and (C) 22.86 m. Orlando strain susceptible females are used as positive controls. Numbers in black above each bar represent the genotyped mosquitoes that survived, and numbers in white within the black zones are the genotyped mosquitoes that died. Different letters (a-c) above bars indicate differences in the proportions of surviving females (Fisher's Exact Test).

https://doi.org/10.1371/journal.pntd.0011033.g004

remained significant even when excluding "distance of 38.1 m" (only for area 23) (S4 Table, Panel A, middle; LR Chi-square, P = 0.0024) and number of females per genotype that were less than 16 (S4 Table Panel A bottom; LR Chi-square, P = 0.0024). The increased "distance" from the sprayer was the most significant factor enhancing survivorship (S4 Table, Panel A, top; LR Chi-square, P < 0.0001), regardless of the exclusion of data at 38.1 m distance (S4 Table, Panel A, middle; LR Chi-square, P < 0.0001) and of genotypes represented by low counts (16 or less females) (S4 Table, Panel A, bottom; LR Chi-square, P < 0.0001). The significant interaction between the 410 site genotype and distance regardless of the consideration of these mentioned exclusions, indicated that the effect of genotype on survivorship will be influenced by distance from the sprayer (S4 Table; LR Chi-square, P < 0.0001).

We tried to measure the effect of genotypes at the 410 site on survival in the context of the genotypes at the other two *kdr* sites, 1016 and 1534 (S4 Table). For this we analyzed the 23 genotypes resulting from the three *kdr* sites (tri-locus genotypes) for 716 females (which include females of area 23 from one test conducted at the distance of 38.1 m instead of 15.24 m). We found there was a significant difference in the survival proportions among the 23 genotypes (Fisher's Exact Test, P = 0.0020; S4 Table). However, the low numbers of certain genotypes detected, such as LL/VV/FF or LL/II/FC each represented by one female (Table 2), prevented to estimate the differences in survivorship between specific tri-locus genotypes (pairwise comparisons could not be performed).

When testing the tri-loci genotypes, distance remained the most significant indicator of survivorship, regardless of excluding data at 38.1 m (S4 Table, Panel B; LR Chi-square, P < 0.0001) and genotypes with less than 16 samples. When all samples and 23 different genotypes were considered, there were significant differences in survivorship (S4 Table, Panel B, top; LR Chi-square, P < 0.0020). When the distance of 38.1 m was removed, we observed that the triple loci genotypes still exhibited differences in survivorship (S4 Table, Panel B, middle; LR Chi-Square, P < 0.0007). However, when excluding the genotypes with less than 16 females, we observed that the 3 most frequent remaining genotypes (LL/II/CC, VL/II/CC, and VV/II/CC) did not appear to have an influence on survivorship (S4 Table, Panel B, bottom; LR Chi-square, P = 0.4240).

The interaction between the tri-locus genotypes and distance when considering all females tested supported that the 23 different genotypes at different distances experienced differences in survivorship (S4 Table, Panel B, top; LR Chi-square, P = 0.0025). When removing the 38.1 m distance, the interaction between genotype and distance remained significant (S4 Table, Panel B, middle; LR Chi-square, P = 0.0009). Upon excluding genotypes and distance interactions with less than 16 samples, the interaction between genotype and distance remained significant (S4 Table, Panel B, bottom; LR Chi-square, P = 0.0002).

Based on multivariate logistic regression, models with the triple mutation genotype (LL/II/ CC as the reference level genotype with higher survivorship), distance (7.62 m as the reference level), area (area 53 as the reference level for higher mortality) and the interaction between "genotype and distance" were significantly associated with survivorship of mosquitoes. Distance again was the most predictive factor for survivorship in the field cage tests, followed by the interaction between distance and genotype (S5 Table). The effects of interaction between genotype and distance, and area effect are visualized using a forest plot (Fig 5). The multiple comparisons supported that genotype of females had little impact on the survival rate at the 7.62 m distance (Fig 5 and S5 Table). At the 15.24- and 22.86-meter distances, females with the triple homozygous resistant mutant LL/II/CC genotype, had a significantly higher survival rate compared to those with VL/II/CC (Fig 5 and S5 Table). When differences by area were considered in the model, we observed that areas 45, 73, 75, and 419 had increased survivorship compared to the reference area 53 but not area 23 (Fig 5 and S5 Table). All pairs of different



Fig 5. Forest plot of the final model showing odds ratio for female survivorship by the interaction of distance and the tri-locus genotypes, and by operational area. Genotypes or areas under <u>A</u> are compared to the genotype or area reference level, indicated by <u>B</u>; the genotype with the highest survivorship, LL/II/CC, and Area 53, with the lowest survivorship, were chosen as the reference levels for this plot.

https://doi.org/10.1371/journal.pntd.0011033.g005

distances are significantly different in survival rate regardless of genotype (<u>S6 Table</u>). That is, survival rate will increase dramatically when distance gets further.

#### Model evaluation

We evaluated the goodness-of-fit of our final model. The McFadden pseudo-R square is 0.54 and area under the curve (AUC) is 93.9%, indicating excellent goodness-of-fit. The LASSO (Least Absolute Shrinkage and Selection Operator) logistic regression model showed that the survivorship predictors with largest coefficients are distance, occurrences of genotype VL/II/ CC at distances of 15.24 or 22.86 m, and area, and therefore validates our final model fitting results. We also performed a sensitivity analysis by fitting the same model using the full data and by fitting the model without "area". The findings from the full data model are consistent with our final model with similar significant terms and similar goodness-of-fit. However, due to the sparsity, the standard errors for many subgroups become extremely large, making the estimate unstable. Therefore, our final model is more reliable compared to full data analysis. A model excluding area has similar model estimates for genotype and distance as well as their interactions, but the goodness-of-fit is poorer with a much smaller pseudo-R square (0.22) and AUC (79.2%), which means our model outperforms this area-excluding model. In short,

although areas 601 and 806 have almost 100% survival rates, adding data of these areas to our final model does not change our conclusions but rather significantly improves the analysis provided by the model.

#### Discussion

The goal of this study was to investigate the effect of the *kdr* V410L mutation, the combined genotypes at 410, 1016 and 1534 sites in the VGSC, and other test variables on mosquito survival after Permanone application. First, we found high frequencies of the V410L *kdr* mutation associated with pyrethroid resistance, manifested as higher than expected survivorship at 7.62 m in most operational areas investigated (Figs 1 and 2). In similar trials with susceptible *Aedes aegypti* 95–100% mortality is expected at 6 m [44].

This is the first detection of the V410L *kdr* mutation in Texas, which was found frequently in combination with the V1016I and F1534C genotypes (Table 2 and Fig 1). In our previous publication we found that while there were high frequencies of the two latter *kdr* alleles, a logistic regression model found no significant association between genotype and survivorship [21], which motivated the current study.

#### Genotype variation among areas of Harris County

Indeed, 81.8% of all females genotyped had at least one resistant allele at the 410 position (Table 2 and Figs 1 and 2). These results match previous studies which had found high frequencies of the V410L genotype in combination with V1016I and F1534C genotypes throughout the world, including the Americas and Africa [20,26–28,45–47].

Of all genotypes detected, the most frequent across all areas in Harris County was the triple homozygous resistant (LL/II/CC) which comprised 48.1% of all mosquitoes tested (Table 2 and S4 Fig). The most common *kdr* alleles in the Americas are 410L + 1016I + 1534C [27]. The presence of the homozygous resistant LL genotype at the 410 site likely contributed to the high female survivorship at the 15.24- and 22.86-m distances from the spray application source (Figs 3 and 4 and S2 and S5 Tables, bottom panel and S6 Table, 2<sup>nd</sup> and 3<sup>rd</sup> panels).

The distribution of V410L genotypes was not even across all areas (Figs 1 and 2). Among the areas analyzed, areas 75, 419, and 45 had the highest proportion of LL females at the 410 site (S1 Fig). For most areas, the proportion of *kdr* genotypes followed a pattern in which the F1534C resistant genotype (CC) was in higher proportion compared to the V1016I resistant genotype (II), which was then followed by the LL genotype at the 410 position (Fig 1). This is in accordance with previous studies that indicate that the V410L mutation has occurred sequentially after the V1016I and 1534C mutations have already been present in a population [27,48]. However, in another study, mosquitoes carrying the V410L mutation were devoid of the F1534C and V1016I genotypes, indicating the V410L mutation can be present independently of the two others [20]. Further, the V410L has been found at higher frequencies compared to the F1534C genes in other countries [49]. In contrast, only in area 53 we found one female of genotype LL/VV/FF and four females VL/VV/FF (Table 2), and in area 73 we found one female of LL/VV/FC. The low frequency of these *kdr* genotypes in Harris County supports a later appearance of the V410L mutation, as was found in the above mentioned studies [27,48].

Of the 23 genotypes found in Harris County for the three *kdr* sites, sixteen carried at least on L allele (Table 2). The L allele appeared most frequently as LL/II/CC genotype the most frequent genotype in each of the areas (48.2%) (Fig 1 and Table 2), followed by the VL/II/CC genotype (19.1%), except for area 75 that only had females of the LL/II/CC genotype. The third most frequent genotype carrying the L mutation was VL/VI/FC (3.8%), although it was not

present in areas 75 and 419. The fourth most common genotype was VL/VI/CC (3.4%), not present in areas 45, 75 and 419. The fifth most common genotype with the L allele was LL/VI/ FC (2.5%), not present in 23, 75 and 419 (Table 2). It should be noted that only 21 mosquitoes were analyzed from area 419. Area 45 was unusual in that it had 24.4% mosquitoes of the LL/ VI/FC genotype, the highest among all areas, which was the second most frequent genotype in that area after LL/II/CC (Table 2). This area supports that the resistant L allele at site 410 could have arisen independently, followed by a crossover event involving established 410L alleles with 1534C alleles, rather than sequential addition of the V1016I mutation to the F1534C, followed by V410L.

#### Influence of V410L genotype on survival in field cage tests

In a previous analysis of these field cage tests, we detected differences in survivorship across areas, distances from spray application, and post positions, but not depending on the combined genotypes at the 1016 and 1534 residue sites when comparing them to susceptible mosquitoes [21]. This led us to investigate if the presence of the V410L mutation may contribute to explain the observed differences in survivorship the *Ae. aegypti* females. When considering the V410L genotypes only, we found that the homozygous resistant LL had higher survivorship compared to the VL and VV genotypes, which had similar survivorship to each other (S2 Fig). This is similar to reports indicating the V410L mutation segregates as a recessive allele [50]. We also found strong linkage disequilibrium between the V1016I and F1534C (S2 Table). Within the same operational area linkage disequilibrium was not identical among the three tested *kdr* loci pairs, which may mean that there are different selection forces in the different areas (S1–S3 Tables).

#### Influence of distance and genotype on the probability of survival

Females from areas 75, 23, and 419 died in the closest cages at 7.62 m (Fig 1), likely overwhelmed by the amount of pesticide received, as this was the distance where the spray volume was the highest [51]. Conversely, the only females tested at 38.1 m were from area 23 and most survived, as there was likely less pesticide that reached that distance (Fig 1) [51]. This is evidenced by studies in which mosquitoes at further distances survived exposure to pyrethroids regardless of genotype [21, 35].

We first analyzed the independent effect of genotypes at the 410 site on mortality achieved at different distances from the pesticide source (S2 Fig); as well as the effect of the tri-locus *kdr* genotypes on mortality at the different distances (Fig 4). The LL genotype at the 410 site significantly increased survivorship at 15.24- and 22.86-m distances from the application source (S2 Fig). Similarly, a positive effect on survivorship for the homozygous resistant mosquitoes with either the 1016 II genotype or the 1534 CC genotype was observed at the 22.86-m distance.

Compared to the susceptible Orlando Strain females, we observed that the three most frequently occurring genotypes in Harris Co. (VV/II/CC, VL/II/CC, and LL/II/CC) had increased survivorship (Fig 4). This is to be expected as the V1016I and F1534C genotypes are associated with pyrethroid resistance in laboratory assays [24]. However, there were differences in survivorship between the triple resistant (LL/II/CC) and VL/II/CC genotypes at all distances (Fig 4) with the LL/II/CC having higher survival at the two further distances (Fig 4B and 4C). Because we observed differences among the three genotypes by distance, we also wanted to determine if area contributed to these differences.

#### Influence of genotype and area on survivorship

Area of origin had a significant negative effect on survivorship for the VL/II/CC genotype only (S4B Fig). These results are mainly driven by area 53, which also had the largest genetic variability, with seventeen of the 23 genotypes (Table 2). We speculate that the lower insecticide selection pressure of only two sprays in the 5 years before this field test, could have decreased other mechanisms of resistance in these mosquitoes (S7 Table). In contrast, area 23 received ten Permanone sprays and area 73, thirteen sprays, and congruent with this higher selection pressure the survivorship of females from both areas was significantly higher than those from area 53 (S7 Table and S4B Fig). For the VL/II/CC genotype the results are similar to a study in California, USA, where area of origin influenced pyrethroid knockdown despite most individual *Ae. aegypti* mosquitoes having identical VGSC *kdr* haplotypes [52]. The area of origin should be interpreted as a covariate that includes the different spraying history, and other selective forces that affect the population genetic background beyond the VGSC genotypes, such as other mechanisms of resistance.

In sum, our results support that area, distance, but genotype only at the 15.24- and 22.86-m distances (Fig 4) significantly affect survivorship when including in the analysis the genotype at the V410L site.

#### Modeling of factors influencing survivorship

Based on the work of Haddi et al. (2017) [26] that demonstrated the V410L genotype is associated with resistance in *Aedes* mosquitoes in the laboratory, several comprehensive analyses were run to determine factors influencing survivorship, including genotypes at the 410 residue site. When first using a logistic regression analysis that included genotypes at 410 position, distance, and the interaction between those two variables, all three factors influenced survivorship (S4 Table, panel A). We then asked if the genotype at the 410 site in conjunction with the genotypes at 1016 and 1534 contributed to survivorship in the field using a second logistic regression model that included "distance" and the "triple locus genotype" and their interaction (S4 Table, Panel B, top).

Similar to the first model, this second model found that survivorship from the field test was influenced by genotype, distance, and the interaction between distance and tri-locus genotype even when excluding the distance of 38.1 m (area 23 only (S4 Table, panel B, top and middle)). However, when excluding genotypes with low counts, leaving only the VV/II/CC, VL/II/CC, and LL/II/CC genotypes for survival comparisons the effect of genotype alone on survivorship could no longer be detected (S4 Table, B, bottom panel). This indicates the influence of these specific genotypes on survival can only be revealed in the field, as less pesticide is received at increased distances and survivorship is differentially affected by genotype.

Although we detected a total of 23 genotypes in Harris County, the high number of genotypes with low number of mosquito representatives (total females in low count genotypes = 190) made them irrelevant in terms of influencing the overall survivorship with respect to the abundant LL/II/CC genotype. The low count genotypes (L.C.G;  $\leq$  16 identified) concentrated in areas 53 (15 L.C.G.), 23 (10 L.C.G.), 73 (9 L.C.G.) and 45 (5 L.C.G.) (Table 2); however, these areas still had a preponderance of the LL/II/CC genotype with 35.9%, 35.9%, 38.5% and 46.75% respectively (Fig 1).

The LASSO analysis selects the most significant factors affecting survivorship. From this analysis a forest plot was derived that yielded the odds of survivorship for genotypes by distance in reference to the tri-locus resistant genotype LL/II/CC, and the odds of survivorship depending on area of origin in relationship to the area with the lowest survivorship (area 53) as reference. This plot showed that for the 15.24- and 22.86-m distances, VL/II/CC genotype

had decreased survivorship compared to the triple resistant LL/II/CC genotype (Fig 5), indicating that two LL alleles are needed for increased survivorship when the mutation is present at the 410 position. Similar to our observations, in laboratory studies mosquitoes with the heterozygous VL genotype at the 410 site, die at higher rates than the LL genotype but have higher recovery after permethrin knockdown. In contrast, we found that although HCPH personnel maintained the treated mosquitoes from the field tests in the laboratory for 24 h and having scored them 10 min after pesticide application, none of the initially "dead" mosquitoes recovered at 24 h. Not all non-synonymous mutations within the VGSC confer resistance to pyrethroids [53]. When considering the analysis of individual *kdr* genotypes at the 410 site on survivorship in the field, it appears that LL contributed to survivorship, and we did not find differences in mortality between VV and VL (Fig 3).

Intriguingly, there was no difference in survivorship at any distance between the homozygous susceptible (VV/II/CC) and homozygous resistant (LL/II/CC) at the 410 site, likely due to the lower representation of VV/II/CC (n = 72) compared to the LL/II/CC (n = 345) when they were further segregated at each distance. This is evident in Fig 4 where VV/II/CC was statistically intermediate in survivorship at the three distances in comparison to the other genotypes, because their lower number in each distance decreased the statistical power to detect differences.

When considering the analysis of individual *kdr* genotypes at the 410 site on survivorship in the field, it appears that LL contributed to survivorship (Fig 3). However, upon analyzing only the genotypes VV/II/CC, VL/II/CC and LL/II/CC simultaneously, the LL genotype does not appear to increase resistance when present with the II/CC genotype (Fig 5 and S4 Table Panel B, bottom). This is congruent with the results of Fig 4, that at both 15.24- and 22.86-m there are no differences in survivorship between the VV/II/CC and VL/II/CC genotypes and that interactions of these two genotypes with distance are not significant (S6 Table, second and third panel, P value VV/II/CC vs VL/II/CC = 0.055 at 15.24-m and P = 0.5746 at 22.86-m). However, the multiple comparisons analyses of the interaction between tri-locus genotypes and distance showed a significant interaction effect, where having an extra L allele (VL/II/CC vs LL/II/CC) at both distances significantly increases the chances of survival (P value at 15.24-m < 0.0001: P value at 22.86-m < 0.0459) (S6 Table). The LL/II/CC genotype is the only genotype for which the interaction and genotype with distance was significant at all distances, indicating that survivorship of this genotype steadily increased with distance (S6 Table, P values for 3 distances each < 0.0001).

Overall, our data support the triple homozygous genotypes appears to have reached a maximum selective advantage at all distances. The results of Fig 3 reflect the fact that the LL genotype is in its majority present in combination with the II/CC genotype.

Other genotype combinations (e.g., VV/VV/FF, LL/VI/FC, etc.) could not be properly analyzed due to low counts of these genotypes, which prevented us from understanding the full effect of each genotype on survival from the field cage tests. Obtaining a congenic strain that only possesses the V410L mutation may provide further insight into the role the V410L genotype plays in resistance. As the mosquitoes used were collected from the field and could not be controlled for genotype, it is likely that there are other resistance mechanisms may be involved.

One mechanism of resistance not analyzed here is metabolic resistance. Overexpression of esterases (ESTs), glutathione s-transferases (GSTs), and cytochrome P450 monooxygenases (P450s), can contribute to pyrethroid resistance [54–56]. The forest plot shows that mosquitoes differed in survivorship based on area of origin (Fig 5). In particular, area 53 had lower survivorship compared to areas 45, 73, 75, and 419 [21]. Notably, area 53 had a significant proportion of dead VL/II/CC mosquitoes compared to other areas (S4 Fig). As the VL mutation is

only weakly resistant to permethrin, and we did not find significant differences in survivorship between VV/II/CC and VL/II/CC genotypes at all distances (<u>S4 Table</u>), it may be that females from area 53 lack other resistance mechanisms. The presence of multiple *kdr* mutations within the VGSC may not be sufficient to determine the level of resistance in populations of *Ae. aegypti.* 

The field cage tests used pyrethroid-susceptible, Orlando-strain females which have been in colony at HCPH-MVCD since 2015, and F<sub>0</sub> females from different operational areas collected as eggs from the field [21]. While cited limitations of using  $F_0$  field-collected females as the treatment group include decreased survivorship due altered rearing conditions in the laboratory [57], in our study all the field-collected control females placed upwind (non-treated), survived after 24-h, the end point of the bioassay, indicating they were fit for the test [21]. Furthermore, females collected from the field as eggs consistently survived Permanone 31-66 exposure at higher frequencies compared to the Orlando mosquitoes. These results supported that the females collected from the field as eggs were viable for use in the field cage tests to assess resistance in Harris County. F<sub>0</sub> females more accurately reflect the genetic diversity present in the field, while the resistant genetic background may be lost during laboratory rearing due to genetic bottlenecks and aggressive breeding selection which can reduce pyrethroid resistance in laboratory colonies in as few as eight generations [58, 59]. However, one challenge of this approach is to obtain a similar number of eggs from all areas. In our study we experienced low representation of females from operational area 23 due to lesser egg collection in the same (Table 2; N = 21). In the laboratory another challenge of F<sub>0</sub> females from field collected eggs required us to synchronize of Orlando-strain female emergence to that of the field collected females [60].

# Conclusions

Most female Ae. aegypti from Harris County were triple resistant for the V410L, V1016I, and F1534C genotypes despite not being targeted for pyrethroid applications by HCPH-MVCD. The high frequency of the LL/II/CC genotype and high survivorship from the field cage tests after exposure to Permanone 31–66 may pose a public health risk in Harris County. As Harris County applies Permanone 31-66 at the highest allowable rate, these results further indicate that "distance" is crucial for "genotype" to influence survivorship in the field with the LL/II/CC genotype having significant positive interactions at all distances. Insecticide concentrations in the field are known to vary depending on application technique, distance from spray, and environmental conditions [61]. Indeed, insecticide applications remain inadequate beyond 7.62-m distances, even here at the highest label rate authorized. A previous study using congenic LL/II/CC females with a susceptible Rockefeller background had similar levels of survivorship to permethrin as F1534C females, but also increased survivorship to deltamethrin [29]. However, these congenic females have not been field-tested as to prove the influence of this genotype under realistic control scenario. The V410L kdr mutation in the Ae. aegypti population of Harris County adds a risk factor to the control of biting females, as LL/II/CC females are associated with increased resistance to deltamethrin, a type-II pyrethroid [29], however are more susceptible to lambda cyhalothrin [54]. We observed that when in combination with the II/CC genotype, females with the heterozygous VL genotype had lower survivorship compared to LL females. This may mean that possession of the VL genotype without other resistance mechanisms may produce a fitness disadvantage for survival in the field tests. Insecticide resistance can remain as long as ten years in Aedes populations without public applications of pyrethroids [62], further complicating control strategies. The high frequencies of the triple resistant genotypes we observed would not be expected unless domestic pesticides are applied

by homeowners and private control companies that would be unaccounted for in this study, which should be considered to avoid the increase of resistance in these populations. Currently, there are no alternatives to pyrethroids that can be used at a wide scale and are approved for public use, aside from malathion which is used in Harris County's rotational strategy for insecticide resistance management. However, malathion is not approved to be applied with a handheld sprayer and concern for cross-resistance remains high.

We have amongst the highest proportion of LL and VL mutations detected and have further supported that possession of resistant genotypes at the 410 site influence of survival from the field cage tests. The selection pressure of pyrethroids in Harris County remains relevant, not only to *Aedes* but *Culex* species in Harris County as well [35]. This resistance poses a threat to chemical-based control strategies in Harris County, highlighting the necessity of novel control strategies for arboviral disease management.

# **Supporting information**

S1 Fig. Percentage of genotypes at the V410L site detected per area across all areas. Across all areas significant differences (Chi-square; P < 0.0001) were detected among the proportion of 410 site genotypes when analyzed independently of genotypes at the 1016 and 1534 sites. Horizontal lines above bars indicate areas in which the percentage of genotypes are not significantly different from pairwise comparisons (Fisher Exact test; P < 0.05). Black numbers above the bars represent the VV genotype, the grey numbers above the bars represent the VL genotype, and the white numbers in the black bars represent the LL genotype. Area 45 is intermediate between the two groups. (TIF)

S2 Fig. Influence of the *kdr* genotype at the 410 site on survivorship of females at each distance after Permanone 31–66 tests. Panels show results of each of the tested distances from the Permanone 31–66 application source, as follows: (A) 7.62 m, (B) 15.24 m, (C) 22.86 m, and (D) 38.1 m. Asterisks above panels indicate there is significantly different survivorship (Fisher's Exact Test; P < 0.05) among genotypes at that distance. Different letters (a-b) above bars indicate differences in the proportions of surviving females using paired comparisons. White numbers in black bars represent the number of dead females and black numbers above the bars represent the number of live females. (TIF)

S3 Fig. Female survival by tri-locus genotype (410, 1016 and 1534) and post position after Permanone 31–66 tests. Post position refers to the placement of the posts within each distance, where post I is the closest to the beginning of the spraying and post III is the one closest to the ending of the spraying. (A) Post position I, (B) Post position II, and (C) Post position III. Numbers above each bar represent the genotyped mosquitoes that survived, and numbers in white within the black zones are the genotyped mosquitoes that died. Different letters (a-b) above bars indicate differences in the proportions of surviving females (Fisher's Exact Test; P < 0.05).

(TIF)

**S4 Fig. Influence of operational area of origin on survivorship of the three most frequent genotypes after Permanone 31–66 tests.** Panels show survival status of the observed geno-types, as follows: (A) LL/II/CC, (B) VL/II/CC, and (C) VV/II/CC. Numbers above each bar represent the genotyped mosquitoes that survived, and numbers in white within the black zones are the genotyped mosquitoes that died. Different letters (a-b) above bars indicate

differences in the proportions of surviving females (Fisher's Exact Test; P < 0.05). (TIF)

S1 Table. Linkage disequilibrium between V410L and V1016I *kdr* genotypes in different operational areas of Harris County. (DOCX)

S2 Table. Linkage disequilibrium between V410L and F1534C *kdr* genotypes in different operational areas of Harris County. (DOCX)

S3 Table. Linkage disequilibrium between V1016 and F1534C *kdr* genotypes in different operational areas of Harris County. (DOCX)

S4 Table. Logistic regression analysis of the genotype at the 410 site (Panel A) and Trilocus genotypes (Panel B) on survivorship. (DOCX)

**S5 Table. Multiple comparisons for the interaction between tri-locus genotype and distance using reference level.** Information provided here was used to develop Fig 5. (\*) Asterisk indicates reference level used for multiple comparisons. Odds ratios are considered significant when the confidence interval of the lower and upper bounds does not contain the value of 1. Number of females used was 526 (see <u>S3 Fig</u> Panel B, bottom). (DOCX)

**S6** Table. Multiple comparisons analysis for the interaction between tri-locus genotype and distance. Bold indicates *P* value is less than or equal to 0.05. (DOCX)

S7 Table. Number of permethrin sprays performed by Harris County Public Health—Mosquito and Vector Control in areas used for field cage tests (FCT) by time before test date. (DOCX)

# Acknowledgments

We thank the A.W.E.S.O.M.E. faculty group of the College of Agriculture and Life Sciences for editing suggestions. We appreciate Drs. Scott Weaver, UTMB, and Phillip E. Kaufman, Texas A&M AgriLife Research, for their Center's leadership.

# **Author Contributions**

Conceptualization: Patricia V. Pietrantonio.

Data curation: Jonathan R. Hernandez.

Formal analysis: Jonathan R. Hernandez, Shuling Liu.

Funding acquisition: Patricia V. Pietrantonio.

Investigation: Patricia V. Pietrantonio.

Methodology: Shuling Liu, Chris L. Fredregill.

Project administration: Patricia V. Pietrantonio.

Resources: Chris L. Fredregill, Patricia V. Pietrantonio.

Supervision: Chris L. Fredregill, Patricia V. Pietrantonio.

Visualization: Jonathan R. Hernandez, Shuling Liu.

Writing - original draft: Jonathan R. Hernandez, Shuling Liu, Patricia V. Pietrantonio.

Writing – review & editing: Jonathan R. Hernandez, Shuling Liu, Chris L. Fredregill, Patricia V. Pietrantonio.

#### References

- Kraemer MU, Sinka ME, Duda KA, Mylne AQ, Shearer FM, Barker CM, et al. The global distribution of the arbovirus vectors *Aedes aegypti* and *Ae. albopictus*. Elife. 2015; 4:e08347. Epub 2015/07/01. https://doi.org/10.7554/eLife.08347 PMID: 26126267
- Naslund J, Ahlm C, Islam K, Evander M, Bucht G, Lwande OW. Emerging mosquito-borne viruses linked to Aedes aegypti and Aedes albopictus: global status and preventive strategies. Vector Borne Zoonotic Dis. 2021; 21(10):731–46. Epub 2021/08/24. https://doi.org/10.1089/vbz.2020.2762 PMID: 34424778
- Brady OJ, Hay SI. The global expansion of dengue: how Aedes aegypti mosquitoes enabled the first pandemic arbovirus. Annu Rev Entomol. 2020; 65(1):191–208. Epub 2019/10/09. https://doi.org/10. 1146/annurev-ento-011019-024918 PMID: 31594415
- Charrel RN, Leparc-Goffart I, Gallian P, de Lamballerie X. Globalization of Chikungunya: 10 years to invade the world. Clin Microbiol Infect. 2014; 20(7):662–3. Epub 2014/06/04. <u>https://doi.org/10.1111/ 1469-0691.12694</u> PMID: 24889861
- McBride CS, Baier F, Omondi AB, Spitzer SA, Lutomiah J, Sang R, et al. Evolution of mosquito preference for humans linked to an odorant receptor. Nature. 2014; 515(7526):222–7. Epub 2014/11/14. https://doi.org/10.1038/nature13964 PMID: 25391959
- Scott TW, Takken W. Feeding strategies of anthropophilic mosquitoes result in increased risk of pathogen transmission. Trends Parasitol. 2012; 28(3):114–21. Epub 2012/02/04. https://doi.org/10.1016/j.pt. 2012.01.001 PMID: 22300806
- World Health Organization. Zika virus outbreaks in the Americas. Weekly Epidemiological Record. 2015; 90(45):609–10.
- Gubler DJ. Epidemic dengue/dengue hemorrhagic fever as a public health, social and economic problem in the 21st century. Trends in Microbiology. 2002; 10(2):100–3. <u>https://doi.org/10.1016/s0966-842x</u> (01)02288-0 PMID: 11827812
- de Curcio JS, Salem-Izacc SM, Pereira Neto LM, Nunes EB, Anunciacao CE, de Paula Silveira-Lacerda E. Detection of Mayaro virus in *Aedes aegypti* mosquitoes circulating in Goiania-Goias-Brazil. Microbes Infect. 2022:104948. Epub 2022/02/03. https://doi.org/10.1016/j.micinf.2022.104948 PMID: 35108606
- 10. Receveur MC, Grandadam M, Pistone T, Malvy D. Infection with Mayaro virus in a French traveller returning from the Amazon region, Brazil, January, 2010. Euro Surveill. 2010; 15(18). Epub 2010/05/13.
- Mweya CN, Kimera SI, Kija JB, Mboera LE. Predicting distribution of Aedes aegypti and Culex pipiens complex, potential vectors of Rift Valley fever virus in relation to disease epidemics in East Africa. Infect Ecol Epidemiol. 2013; 3. Epub 2013/10/19. https://doi.org/10.3402/iee.v3i0.21748 PMID: 24137533
- Espinal MA, Andrus JK, Jauregui B, Waterman SH, Morens DM, Santos JI, et al. Emerging and reemerging *Aedes*-transmitted arbovirus infections in the region of the Americas: implications for health policy. Am J Public Health. 2019; 109(3):387–92. Epub 2019/01/25. https://doi.org/10.2105/AJPH. 2018.304849 PMID: 30676796
- Narahashi T. Neuronal ion channels as the target sites of insecticides. Pharmacol Toxicol. 1996; 79 (1):1–14. Epub 1996/07/01. https://doi.org/10.1111/j.1600-0773.1996.tb00234.x PMID: 8841090.
- Flores AE, Ponce G, Silva BG, Gutierrez SM, Bobadilla C, Lopez B, et al. Wide spread cross resistance to pyrethroids in *Aedes aegypti* (Diptera: Culicidae) from Veracruz state Mexico. J Econ Entomol. 2013; 106(2):959–69. Epub 2013/06/22. https://doi.org/10.1603/ec12284 PMID: 23786088
- Moyes CL, Vontas J, Martins AJ, Ng LC, Koou SY, Dusfour I, et al. Contemporary status of insecticide resistance in the major *Aedes* vectors of arboviruses infecting humans. PLoS Negl Trop Dis. 2017; 11 (7):e0005625. Epub 2017/07/21. https://doi.org/10.1371/journal.pntd.0005625 PMID: 28727779
- Smith LB, Kasai S, Scott JG. Pyrethroid resistance in *Aedes aegypti* and *Aedes albopictus*: Important mosquito vectors of human diseases. Pestic Biochem Physiol. 2016; 133:1–12. Epub 2016/10/16. https://doi.org/10.1016/j.pestbp.2016.03.005 PMID: 27742355.
- Williamson MS, Martinez-Torres D, Hick CA, Devonshire AL. Identification of mutations in the housefly para-type sodium channel gene associated with knockdown resistance (kdr) to pyrethroid insecticides.

Mol Gen Genet. 1996; 252(1–2):51–60. Epub 1996/08/27. https://doi.org/10.1007/BF02173204 PMID: 8804403.

- Chen M, Du Y, Nomura Y, Zhorov BS, Dong K. Chronology of sodium channel mutations associated with pyrethroid resistance in *Aedes aegypti*. Arch Insect Biochem Physiol. 2020; 104(2):e21686. Epub 2020/05/08. https://doi.org/10.1002/arch.21686 PMID: 32378259.
- Cosme LV, Gloria-Soria A, Caccone A, Powell JR, Martins AJ. Evolution of *kdr* haplotypes in worldwide populations of *Aedes aegypti*: Independent origins of the F1534C *kdr* mutation. PLoS Negl Trop Dis. 2020; 14(4):e0008219. Epub 2020/04/17. https://doi.org/10.1371/journal.pntd.0008219 PMID: 32298261.
- Saavedra-Rodriguez K, Maloof FV, Campbell CL, Garcia-Rejon J, Lenhart A, Penilla P, et al. Parallel evolution of vgsc mutations at domains IS6, IIS6 and IIIS6 in pyrethroid resistant Aedes aegypti from Mexico. Sci Rep. 2018; 8(1):6747. Epub 2018/05/02. <u>https://doi.org/10.1038/s41598-018-25222-0</u> PMID: 29712956.
- Hernandez JR, Longnecker M, Fredregill CL, Debboun M, Pietrantonio PV. Kdr genotyping (V1016I, F1534C) of the Nav channel of Aedes aegypti (L.) mosquito populations in Harris County (Houston), Texas, USA, after Permanone 31–66 field tests and its influence on probability of survival. PLoS Negl Trop Dis. 2021; 15(11):e0009833. Epub 2021/11/05. https://doi.org/10.1371/journal.pntd.0009833 PMID: 34735439.
- Fan Y, Scott JG. The F1534C voltage-sensitive sodium channel mutation confers 7- to 16-fold resistance to pyrethroid insecticides in *Aedes aegypti*. Pest Manag Sci. 2020; 76(6):2251–9. Epub 2020/01/26. https://doi.org/10.1002/ps.5763 PMID: 31981401.
- Du Y, Nomura Y, Satar G, Hu Z, Nauen R, He SY, et al. Molecular evidence for dual pyrethroid-receptor sites on a mosquito sodium channel. Proc Natl Acad Sci U S A. 2013; 110(29):11785–90. Epub 2013/ 07/04. https://doi.org/10.1073/pnas.1305118110 PMID: 23821746.
- Chen M, Du Y, Wu S, Nomura Y, Zhu G, Zhorov BS, et al. Molecular evidence of sequential evolution of DDT- and pyrethroid-resistant sodium channel in *Aedes aegypti*. PLoS Negl Trop Dis. 2019; 13(6): e0007432. Epub 2019/06/04. https://doi.org/10.1371/journal.pntd.0007432 PMID: 31158225.
- 25. Plernsub S, Saingamsook J, Yanola J, Lumjuan N, Tippawangkosol P, Sukontason K, et al. Additive effect of knockdown resistance mutations, S989P, V1016G and F1534C, in a heterozygous genotype conferring pyrethroid resistance in *Aedes aegypti* in Thailand. Parasit Vectors. 2016; 9(1):417. Epub 2016/07/28. https://doi.org/10.1186/s13071-016-1713-0 PMID: 27460671.
- Haddi K, Tome HVV, Du Y, Valbon WR, Nomura Y, Martins GF, et al. Detection of a new pyrethroid resistance mutation (V410L) in the sodium channel of *Aedes aegypti*: a potential challenge for mosquito control. Sci Rep. 2017; 7:46549. Epub 2017/04/20. https://doi.org/10.1038/srep46549 PMID: 28422157.
- Fan Y, O'Grady P, Yoshimizu M, Ponlawat A, Kaufman PE, Scott JG. Evidence for both sequential mutations and recombination in the evolution of *kdr* alleles in *Aedes aegypti*. PLoS Negl Trop Dis. 2020; 14(4):e0008154. Epub 2020/04/18. https://doi.org/10.1371/journal.pntd.0008154 PMID: 32302303.
- Villanueva-Segura OK, Ontiveros-Zapata KA, Lopez-Monroy B, Ponce-Garcia G, Gutierrez-Rodriguez SM, Davila-Barboza JA, et al. Distribution and frequency of the kdr mutation V410L in natural populations of *Aedes aegypti* (L.) (Diptera: Culicidae) from eastern and southern Mexico. J Med Entomol. 2020; 57(1):218–23. Epub 2019/09/11. https://doi.org/10.1093/jme/tjz148 PMID: 31504686.
- Silva JJ, Kouam CN, Scott JG. Levels of cross-resistance to pyrethroids conferred by the Vssc knockdown resistance allele 410L+1016l+1534C in *Aedes aegypti*. PLoS Negl Trop Dis. 2021; 15(7): e0009549. Epub 2021/07/13. https://doi.org/10.1371/journal.pntd.0009549 PMID: 34252110.
- U.S. Census Bureau. United States Census Bureau QuickFacts 2020. <u>https://www.census.gov/</u> quickfacts/harriscountytexas.
- **31.** Mukherjee F, Singh D. Detecting flood prone areas in Harris County: a GIS based analysis. GeoJournal. 2019; 85(3):647–63. https://doi.org/10.1007/s10708-019-09984-2
- 32. Reina Ortiz M, Le NK, Sharma V, Hoare I, Quizhpe E, Teran E, et al. Post-earthquake Zika virus surge: Disaster and public health threat amid climatic conduciveness. Sci Rep. 2017; 7(1):15408. Epub 2017/ 11/15. https://doi.org/10.1038/s41598-017-15706-w PMID: 29133932.
- Harris JW, Richards SL, Anderson A. Emergency mosquito control on a selected area in eastern north Carolina after hurricane Irene. Environ Health Insights. 2014; 8(Suppl 2):29–33. Epub 2015/01/13. https://doi.org/10.4137/EHI.S16001 PMID: 25574141.
- Vigilant M, Battle-Freeman C, Braumuller KC, Riley R, Fredregill CL. Harris County Public Health Mosquito and Vector Control Division emergency response to Hurricane Harvey: vector-borne disease surveillance and control. J Am Mosq Control Assoc. 2020; 36(2s):15–27. Epub 2021/03/02. <u>https://doi.org/ 10.2987/19-6890.1 PMID: 33647149</u>.
- **35.** Lee HJ, Longnecker M, Calkins TL, Renfro AD, Fredregill CL, Debboun M, et al. Detection of the Nav channel *kdr*-like mutation and modeling of factors affecting survivorship of *Culex quinquefasciatus*

mosquitoes from six areas of Harris County (Houston), Texas, after permethrin field-cage tests. PLoS Negl Trop Dis. 2020; 14(11):e0008860. Epub 2020/11/20. https://doi.org/10.1371/journal.pntd.0008860 PMID: 33211688.

- 36. London Fog. Colt-4 ULV Handheld Aerosol Generator Manual. Long Lake, Minnesota. 2020.
- Wilke ABB, Vasquez C, Carvajal A, Moreno M, Fuller DO, Cardenas G, et al. Urbanization favors the proliferation of *Aedes aegypti* and *Culex quinquefasciatus* in urban areas of Miami-Dade County, Florida. Sci Rep. 2021; 11(1):22989. Epub 2021/11/28. <u>https://doi.org/10.1038/s41598-021-02061-0</u> PMID: 34836970.
- Pareja-Loaiza PX, Santacoloma Varon L, Rey Vega G, Gomez-Camargo D, Maestre-Serrano R, Lenhart A. Mechanisms associated with pyrethroid resistance in populations of *Aedes aegypti* (Diptera: Culicidae) from the Caribbean coast of Colombia. PLoS One. 2020; 15(10):e0228695. Epub 2020/10/ 07. https://doi.org/10.1371/journal.pone.0228695 PMID: 33022007.
- Estep AS, Sanscrainte ND, Waits CM, Bernard SJ, Lloyd AM, Lucas KJ, et al. Quantification of permethrin resistance and kdr alleles in Florida strains of *Aedes aegypti* (L.) and *Aedes albopictus* (Skuse). PLoS Negl Trop Dis. 2018; 12(10):e0006544. Epub 2018/10/26. https://doi.org/10.1371/journal.pntd. 0006544 PMID: 30356237.
- Estep AS, Sanscrainte ND, Waits CM, Louton JE, Becnel JJ. Resistance status and resistance mechanisms in a strain of *Aedes aegypti* (Diptera: Culicidae) from Puerto Rico. J Med Entomol. 2017; 54 (6):1643–8. Epub 2017/10/06. https://doi.org/10.1093/jme/tjx143 PMID: 28981681.
- Abbott WS. A method of computing the effectiveness of an insecticide. J Econ Entomol. 1925; 18 (2):265–7. https://doi.org/10.1093/jee/18.2.265a
- Okimoto R, Dodgson JB. Improved PCR amplification of multiple specific alleles (PAMSA) using internally mismatched primers. Biotechniques. 1996; 21(1):20–6. https://doi.org/10.2144/96211bm03 PMID: 8816228
- **43.** Tibshirani R. Regression shrinkage and selection via the Lasso. J Roy Stat Soc B Met. 1996; 58 (1):267–88.
- Harburguer L, Seccacini E, Licastro S, Zerba E, Masuh H. Droplet size and efficacy of an adulticide-larvicide ultralow-volume formulation on *Aedes aegypti* using different solvents and spray application methods. Pest Manag Sci. 2012; 68(1):137–41. Epub 2011/06/30. https://doi.org/10.1002/ps.2237 PMID: 21714060.
- 45. Itokawa K, Furutani S, Takaoka A, Maekawa Y, Sawabe K, Komagata O, et al. A first, naturally occurring substitution at the second pyrethroid receptor of voltage-gated sodium channel of *Aedes aegypti*. Pest Manag Sci. 2021; 77(6):2887–93. Epub 2021/02/10. https://doi.org/10.1002/ps.6324 PMID: 33559956.
- 46. Ayres CFJ, Seixas G, Borrego S, Marques C, Monteiro I, Marques CS, et al. The V410L knockdown resistance mutation occurs in island and continental populations of *Aedes aegypti* in West and Central Africa. PLoS Negl Trop Dis. 2020; 14(5):e0008216. Epub 2020/05/10. <u>https://doi.org/10.1371/journal.pntd.0008216</u> PMID: 32384079.
- 47. Villanueva-Segura K, Ponce-Garcia G, Lopez-Monroy B, Mora-Jasso E, Perales L, Gonzalez-Santillan FJ, et al. Multiplex PCR for simultaneous genotyping of *kdr* mutations V410L, V1016l and F1534C in *Aedes aegypti* (L.). Parasit Vectors. 2020; 13(1):325. Epub 2020/06/27. https://doi.org/10.1186/s13071-020-04193-0 PMID: 32586378.
- Vera-Maloof FZ, Saavedra-Rodriguez K, Elizondo-Quiroga AE, Lozano-Fuentes S, Black Iv WC. Coevolution of the ile1,016 and cys1,534 mutations in the voltage gated sodium channel gene of *Aedes* aegypti in Mexico. PLoS Negl Trop Dis. 2015; 9(12):e0004263. Epub 2015/12/15. <u>https://doi.org/10. 1371/journal.pntd.0004263</u> PMID: 26658798.
- Sintya E, Sari K., Widhidewi N. W., Sukmawati N. M., Witari N. P., & Widarsa T. Hubungan frekuensi gen knock-down resistance (*kdr*) V1016G, V410L, dan F1534C dengan tingkat resistensi populasi *Aedes aegypti* di Denpasar, Bali. [In Indonesian]. Jurnal Vektor Penyakit. 2022; 15:73–82. https://doi. org/10.22435/vektorp.v15i2.4907
- Saavedra-Rodriguez K, Campbell CL, Lozano S, Penilla-Navarro P, Lopez-Solis A, Solis-Santoyo F, et al. Permethrin resistance in *Aedes aegypti*: Genomic variants that confer knockdown resistance, recovery, and death. PLoS Genet. 2021; 17(6):e1009606. Epub 2021/06/18. https://doi.org/10.1371/ journal.pgen.1009606 PMID: 34138859.
- Farooq M, Salyani M, Walker TW. Droplet characteristics and near nozzle dispersion of cold and thermal fog. Pesticide formulation and delivery systems: 32nd volume, innovating legacy products for new uses. 2013. p. 49–64.
- 52. Mack LK, Kelly ET, Lee Y, Brisco KK, Shen KV, Zahid A, et al. Frequency of sodium channel genotypes and association with pyrethrum knockdown time in populations of Californian *Aedes aegypti*. Parasit

Vectors. 2021; 14(1):141. Epub 2021/03/08. https://doi.org/10.1186/s13071-021-04627-3 PMID: 33676552.

- Silva JJ, Scott JG. Conservation of the voltage-sensitive sodium channel protein within the Insecta. Insect Mol Biol. 2020; 29(1):9–18. Epub 2019/06/18. <u>https://doi.org/10.1111/imb.12605</u> PMID: 31206812.
- 54. Granada Y, Mejia-Jaramillo AM, Zuluaga S, Triana-Chavez O. Molecular surveillance of resistance to pyrethroids insecticides in Colombian Aedes aegypti populations. PLoS Negl Trop Dis. 2021; 15(12): e0010001. Epub 2021/12/15. https://doi.org/10.1371/journal.pntd.0010001 PMID: 34905537.
- Schluep SM, Buckner EA. Metabolic resistance in permethrin-resistant Florida Aedes aegypti (Diptera: Culicidae). Insects. 2021; 12(10). Epub 2021/10/24. https://doi.org/10.3390/insects12100866 PMID: 34680634.
- Marcombe S, Mathieu RB, Pocquet N, Riaz MA, Poupardin R, Selior S, et al. Insecticide resistance in the dengue vector *Aedes aegypti* from Martinique: distribution, mechanisms and relations with environmental factors. PLoS One. 2012; 7(2):e30989. Epub 2012/03/01. <u>https://doi.org/10.1371/journal.pone.</u> 0030989 PMID: 22363529.
- Xu T, Zhong D, Tang L, Chang X, Fu F, Yan G, et al. *Anopheles sinensis* mosquito insecticide resistance: comparison of three mosquito sample collection and preparation methods and mosquito age in resistance measurements. Parasit Vectors. 2014; 7:54. Epub 2014/01/30. <u>https://doi.org/10.1186/1756-3305-7-54 PMID: 24472598</u>.
- Vera-Maloof FZ, Saavedra-Rodriguez K, Penilla-Navarro RP, A DR-R, Dzul F, Manrique-Saide P, et al. Loss of pyrethroid resistance in newly established laboratory colonies of *Aedes aegypti*. PLoS Negl Trop Dis. 2020; 14(3):e0007753. Epub 2020/03/17. https://doi.org/10.1371/journal.pntd.0007753 PMID: 32176693.
- Ffrench-Constant RH. The molecular genetics of insecticide resistance. Genetics. 2013; 194(4):807– 15. Epub 2013/08/03. https://doi.org/10.1534/genetics.112.141895 PMID: 23908373.
- Rajatileka S, Burhani J, Ranson H. Mosquito age and susceptibility to insecticides. Trans R Soc Trop Med Hyg. 2011; 105(5):247–53. Epub 2011/03/01. <u>https://doi.org/10.1016/j.trstmh.2011.01.009</u> PMID: 21353689.
- Ebert TA, Taylor RAJ, Downer RA, Hall FR. Deposit structure and efficacy of pesticide application. 1: Interactions between deposit size, toxicant concentration and deposit number. Pesticide Science. 1999; 55(8):783–92.
- Macoris ML, Martins AJ, Andrighetti MTM, Lima JBP, Valle D. Pyrethroid resistance persists after ten years without usage against *Aedes aegypti* in governmental campaigns: Lessons from Sao Paulo State, Brazil. PLoS Negl Trop Dis. 2018; 12(3):e0006390. Epub 2018/03/31. https://doi.org/10.1371/ journal.pntd.0006390 PMID: 29601580.